Author’s response to reviews

Title: Evaluation of variants in the selectin genes in age-related macular degeneration.

Authors:

Robert F Mullins (robert-mullins@uiowa.edu)
Jessica M Skeie (jessica-skeie@uiowa.edu)
James C Folk (james-Folk@uiowa.edu)
Frances M Solivan-Timpe (frances-solivan@uiowa.edu)
Thomas A Oetting (thomas-oetting@uiowa.edu)
Jian Huang (jian-huang@uiowa.edu)
Kai Wang (kai-wang@uiowa.edu)
Edwin M Stone (edwin-stone@uiowa.edu)
John H Fingert (john-fingert@uiowa.edu)

Version: 4 Date: 6 April 2011

Author’s response to reviews: see over
Dear BMC Editors,

Thank you for your assistance with our manuscript, “Evaluation of variants in the selectin genes in age-related macular degeneration.”

We have modified the MS based on your suggestions as described below:

1. Affiliations have been added for all authors.
2. A Conclusion section has been added
3. Figure titles and legends have been added to the end of the MS.
4. We have now included the tables into a tables section in the MS
5. We have changed the case of the title characters in the MS
6. We have removed highlighting from the MS, with the exception of Tables 1 and 2 in which the highlighting indicates SNPs that were statistically significant
7. We have uploaded the figures as jpg files

Thank you for your help.

Best wishes,

John H Fingert, MD, PhD
Department of Ophthalmology and Visual Sciences
The University of Iowa